Author's response to reviews

Title: Progressive multifocal leucoencephalopathy in an immunocompetent patient with favourable outcome. A case report.

Authors:

Halvor Naess (haln@haukeland.no)
Solveig Glad (sgla@haukeland.no)
Anette Storstein (astt@haukeland.no)
Christine H Rinaldo (Christine.Hanssen.Rinaldo@unn.no)
Sverre J Mørk (svjm@haukeland.no)
Kjell M Myhr (kimy@haukeland.no)
Hans Hirsch (Hans.Hirsch@unibas.ch)

Version: 3 Date: 13 May 2010

Author's response to reviews:

1. The abstract is changed as requested.
2. The author list is changed as requested.
3. References are changed as requested.
4. Consent, competing interests and authors contribution has been moved before references.
5. The format of authors contribution has been changed as requested.
6. Figure titles have been provided.
7. Format of authors email adress has been changed.
8. References are changed as requested.
9. Figure names have been removed from the image files.
10. White space has been removed from Figure 1, but I was unable remove white space from Figure 2.